Author's response to reviews

Title: Assisted reproductive technologies (ARTs): Evaluation of evidence to support public policy development

Authors:

Alexa A Nardelli (alexan@ualberta.ca)
Tania Stafinski (tialias@ualberta.ca)
Tarek Motan (tmo@ualberta.ca)
Kristin Klein (kristinklein@ualberta.ca)
Devidas Menon (menon@ualberta.ca)

Version: 4 Date: 6 October 2014

Author's response to reviews:

Please find attached a response to the reviewers' comments regarding the manuscript entitled “Assisted reproductive technologies (ARTs): Evaluation of evidence to support public policy development” (previous title: “A Systematic Review of the Safety and Effectiveness of In Vitro Fertilization: the impact of procedural factors”), authored by myself, Tania Stafinski, Tarek Motan, Kristin Klein, and Devidas Menon.